Author's response to reviews

Title: Can we rely on the best trial? A comparison of individual trials and systematic reviews

Authors:

Paul P Glasziou (paul.glasziou@dphpc.ox.ac.uk)
Sasha Shepperd (Sasha.Shepperd@dphpc.ox.ac.uk)
Jon Brassey (jon@tripdatabase.com)

Version: 4  Date: 4 March 2010

Author's response to reviews:

Dear Editors

Ask requested by the editors we have made the following changes:

1. Corrected the reference numbering.
2. Removed the Header
3. Removed the Figure and Table placement text
4. Removed the vertical lines from the Tables.
5. Checked for other errors (and have amended the header for one table).